Author's response to reviews

Title: Chronic hepatitis caused by persistent parvovirus B19 infection

Authors:

Trine H Mogensen (trine.mogensen@dadlnet.dk)
Jens Magnus B Jensen (iejensen@rm.dk)
Stephen Hamilton-Dutoit (stephami@rm.dk)
Carsten S Larsen (carslars@rm.dk)

Version: 4 Date: 14 August 2010

Author's response to reviews:

Dear Editor,

Thank you for your response and acceptance (in principle) of our manuscript. We here send a revised version of the manuscript in which we have (1) added full details of author affiliations, including email addresses, on the title page, and (2) have moved the statement about consent to the Acknowledgments section. We believe the manuscript conforms to the journal style, and hope you can now formally accept our manuscript in BMC Infectious Diseases.

Yours sincerely,
Trine Mogensen